Author's response to reviews

Title: Integration of rural and urban medical insurance schemes in China: an empirical research

Authors:

Xin Wang (zhengyuang1993@yahoo.com.cn)
Ang Zheng (Angzheng1993@yahoo.cn)
Xin He (695436829@qq.com)
Hanghang Jiang (497363047@qq.com)

Version: 2 Date: 9 November 2012

Author's response to reviews:

I have made the following revisions:
1. I have added the title page in the manuscript;
2. a 'Competing interests' section was added in the manuscript;
3. an Authors’ Contributions section after Competing interests was included in the manuscript;
4. ethical approval was stated in the manuscript;
5. written informed consent was stated in the manuscript;
6. the manuscript was formatted according to the guideline.